Author's response to reviews

Title: Detection of infectious organisms in archival prostate cancer tissues

Authors:

Melissa A Yow (melissa.yow@gmail.com)
Sepehr N Tabrizi (sepehr@unimelb.edu.au)
Gianluca Severi (gianluca.severi@hugef-torino.org)
Damien M Bolton (damienmb@unimelb.edu.au)
John Pedersen (john.pedersen@tissupath.com.au)
Anthony Longano (Anthony.Longano@southernhealth.org.au)
Suzanne M Garland (suzanne.garland@rch.org.au)
Melissa C Southey (msouthey@unimelb.edu.au)
Graham G Giles (Graham.giles@cancervic.org.au)

Version: 5
Date: 21 July 2014

Author's response to reviews: see over
Authors’ response to editorial requests

MS: 7563623591272260

Editorial requests:
--------------
- Please could you clarify whether you have ethics committee approval for the use of these samples and please provide the full name of the ethics committee’s in your methods section.
  
  • The following text has been added to the methods section: “Approval for use of the samples arising from these studies was given by the Human Research Ethics Committee of Cancer Council Victoria.”

- Please include email addresses for all authors on your title page.
  
  • The email addresses of all authors have been added to the title page.